Author’s response to reviews

Title: Fatal accidental methanol ingestion

Authors:

Jelle L Epker (j.epker@erasmusmc.nl)
Jan Bakker (jan.bakker@erasmusmc.nl)

Version: 8 Date: 8 February 2010

Author’s response to reviews:

Dear Madam, Sir,

As suggested I changed the abstract structure to: background, case description and conclusions.

The same was done with the case report itself. The heading "methods" was removed as was the "discussion" heading. The discussion is as suggested in the instruction for authors part of the case description (no subheadings were added, though they might be appropriate?)

I do hope that everything is now as it should be, best wishes,

Jelle Epker